

Citation: Onken A, Moyo S, Miraji MK, Bohlin J, Marijani M, Manyahi J, et al. (2024) Predominance of multidrug-resistant *Salmonella* Typhi genotype 4.3.1 with low-level ciprofloxacin resistance in Zanzibar. PLoS Negl Trop Dis 18(4): e0012132. https://doi.org/10.1371/journal.pntd.0012132

Editor: Joseph M. Vinetz, Yale University School of Medicine, UNITED STATES

Received: October 6, 2023

Accepted: April 2, 2024

Published: April 17, 2024

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0012132

Copyright: © 2024 Onken et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The whole genome sequencing data for this study (fastq files of all short read sequencing analyzes and five of the long read assemblies of the nine selected strains) have

RESEARCH ARTICLE

Predominance of multidrug-resistant *Salmonella* Typhi genotype 4.3.1 with low-level ciprofloxacin resistance in Zanzibar

Annette Onken^{1,2,3}*, Sabrina Moyo^{1,4}, Mohammed Khamis Miraji⁵, Jon Bohlin^{6,7}, Msafiri Marijani⁸, Joel Manyahi⁹, Kibwana Omar Kibwana⁹, Fredrik Müller^{10,11}, Pål A. Jenum^{3,11}, Khamis Ali Abeid¹², Marianne Reimers¹³, Nina Langeland^{1,2}, Kristine Mørch^{1,2}, Bjørn Blomberg^{1,2}

 Department of Clinical Science, University of Medicine, Bergen, Norway, 2 National Centre for Tropical Infectious Diseases, Haukeland University Hospital, Bergen, Norway, 3 Department of Microbiology, Vestre Viken Hospital Trust, Drammen, Norway, 4 Department of Tropical Disease Biology, Liverpool School of Tropical Medicine, Liverpool, United Kingdom, 5 Department of Internal Medicine, Mnazi Mmoja Hospital, Zanzibar, Tanzania, 6 Department of methods and analysis, Section of modelling and bioinformatics, Domain of Infection Control, Oslo, Norway, 7 Center for Fertility and Health analysis, Norwegian Institute of Public Health, Oslo, Norway, 8 Pathology Laboratory Department, Mnazi Mmoja Hospital, Zanzibar, Tanzania,
Department of Microbiology and Immunology, Muhimbili University of Health and Allied Sciences, Dar es Salaam, Tanzania, 10 Department of Microbiology, Oslo University Hospital, Oslo, Norway, 11 Institute of Clinical Medicine, University of Oslo, Oslo, Norway, 12 Department of Pediatrics, Mnazi Mmoja Hospital, Zanzibar, Tanzania, 13 Emergency Care Clinic, Haukeland University Hospital, Bergen, Norway

* annetteonken@yahoo.com

Abstract

Background

Typhoid fever is a common cause of febrile illness in low- and middle-income countries. While multidrug-resistant (MDR) *Salmonella* Typhi (*S*. Typhi) has spread globally, fluoroquinolone resistance has mainly affected Asia.

Methods

Consecutively, 1038 blood cultures were obtained from patients of all age groups with fever and/or suspicion of serious systemic infection admitted at Mnazi Mmoja Hospital, Zanzibar in 2015–2016. *S.* Typhi were analyzed with antimicrobial susceptibility testing and with short read (61 strains) and long read (9 strains) whole genome sequencing, including three *S.* Typhi strains isolated in a pilot study 2012–2013.

Results

Sixty-three *S*. Typhi isolates (98%) were MDR carrying bla_{TEM-1B} , *sul1* and *sul2*, *dfrA7* and *catA1* genes. Low-level ciprofloxacin resistance was detected in 69% (43/62), with a single gyrase mutation *gyrA*-D87G in 41 strains, and a single *gyrA*-S83F mutation in the non-MDR strain. All isolates were susceptible to ceftriaxone and azithromycin. All MDR isolates belonged to genotype 4.3.1 lineage I (4.3.1.1), with the antimicrobial resistance

been deposited in the European Nucleotide Archive (ENA) at EMBL-EBI under accession number PRJEB59168 (https://www.ebi.ac.uk/ena/browser/ view/PRJEB59168). The remaining four long read assemblies have been deposited in GenBank, National Institutes of Health (NIH), BioProject number PRJNA982791 (GenBank Overview (nih. gov)), accession numbers SAMN35714885 (ZNZ13L78), SAMN35713968 (ZNZ17F60), SAMN35714917 (ZNZ55M142), and SAMN35714939 (ZNZ57M158).). All other relevant data are within the manuscript, with the accession numbers of the genomes and additional information listed in the Supplementary information (S1 Table).

Funding: The study received funding from the Norwegian National Centre on Tropical Infectious Diseases, Haukeland University Hospital, Bergen, Norway. Vestre Viken Hospital Trust, Drammen, Norway supported the study including a salary to AO. Western Norway Regional Health Authority, University of Bergen, Norway supported AO by a PhD scholarship (Project Number 912277). CAMRIA - Combatting Anti-Microbial Resistance with Interdisciplinary Approaches, Centre for Antimicrobial Resistance in Western Norway, funded by Trond Mohn Foundation, supported the study (grant number TMS2020TMT11). SM has received a salary from STRESST - Antimicrobial Stewardship in Hospitals, Resistance Selection and Transfer in a One Health Context, University of Bergen, Norway (JPIAMR grant number NFR333432). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

determinants located on a composite transposon integrated into the chromosome. Phylogenetically, the MDR subgroup with ciprofloxacin resistance clusters together with two external isolates.

Conclusions

We report a high rate of MDR and low-level ciprofloxacin resistant *S*. Typhi circulating in Zanzibar, belonging to genotype 4.3.1.1, which is widespread in Southeast Asia and African countries and associated with low-level ciprofloxacin resistance. Few therapeutic options are available for treatment of typhoid fever in the study setting. Surveillance of the prevalence, spread and antimicrobial susceptibility of *S*. Typhi can guide treatment and control efforts.

Author summary

Salmonella Typhi causes typhoid fever. Multi-drug resistant (MDR) S. Typhi is spreading globally. Local and regional surveillance of MDR S. Typhi populations using both blood culture and whole genome sequencing can uncover outbreaks and help mapping the spread of S. Typhi and resistance mechanisms, which, in turn, can guide both control and prevention efforts and clinical management. Data regarding the distribution of MDR S. Typhi genotypes and resistance mechanisms is scarce in Zanzibar, Tanzania, as in many other African countries. In this study we characterize S. Typhi phenotypically and genotypically. This study shows a high rate of MDR S. Typhi, hence few therapeutic options are available for treatment of typhoid fever in the study setting. Our findings contribute to the knowledge base on typhoid fever in the region and to guide correct treatment of individual patients and control of the disease.

Introduction

Salmonella enterica subspecies enterica serovar Typhi (S. Typhi) causes typhoid fever, an important global health problem with an estimated worldwide annual incidence of over 9 million cases, and about 100,000 to 200,000 deaths [1-4]. The estimated burden is uncertain due to diagnostic shortcomings and poor access to health care [5], but there are indications that the burden has declined lately [2]. In different geographic regions the incidence varies substantially, with incidence in excess of 800 cases per 100 000 persons per year in some settings in sub-Saharan Africa [6,7], and higher numbers in certain urban areas compared to rural settings [7]. Typhoid fever prevails in regions with poor sanitation facilities and limited access to clean drinking water [8]. Infections caused by multidrug-resistant (MDR) S. Typhi strains, defined by resistance to the prior first-line treatments ampicillin, chloramphenicol and trimethoprim-sulfamethoxazole, are complicated to treat and are associated with increased mortality [9,10]. The appearance of MDR S. Typhi in the 1970s led to widespread use of fluoroquinolones, and subsequent emergence of fluoroquinolone resistant S. Typhi in the early 1990s [11–13]. A combination of MDR and fluoroquinolone resistant S. Typhi leaves clinicians with few therapeutic options especially in developing countries. Since 2017, WHO ranked fluoroquinolone resistant S. Typhi as a high priority pathogen for research and development of new antibiotics [14].

Genetically S. Typhi is monomorphic [13]. A recent introduction of a phylogenetic genotyping scheme, GenoTyphi, has facilitated the interpretation of whole genome sequencing data (WGS) for S. Typhi [15–17]. Using this scheme, a global collection of S. Typhi isolates showed that the S. Typhi population is comprised of dozens of subclades which are specific for different geographical locations [15,17,18]. These global surveillance studies have shown that the majority of MDR S. Typhi infections worldwide belong to genotype 4.3.1 [18], which originated from South Asia and spread up to East Africa [15,17–19].

MDR in *S*. Typhi is linked to the presence of a composite transposon. In the 4.3.1. genotype, it was first introduced via the IncHI1-PST6 plasmid [18–21]. Later, reports have shown that the transposon carrying the genes associated with MDR was integrated into the *S*. Typhi chromosome [18,22,23]. This composite transposon carries antimicrobial resistance determinants which confer resistance towards penicillins (*bla*_{TEM-1}), trimethoprim (*dfrA7*) and sulfon-amides (*sul1*, *sul2*), chloramphenicol (*catA1*) and to streptomycin (*strA*, *strB*) [18,21,23].

Resistance against fluoroquinolones can result from mutations in the quinolone-resistancedetermining regions (QRDRs) of the chromosomal *gyr* and *par* genes [10,24,25] and/or plasmid-mediated quinolone resistance (PMQR) which involves acquisition of e.g. *qnr* genes [24,26–28]. Using WGS, a large study [18] reported that the global population of MDR *S*. Typhi with reduced susceptibility to fluoroquinolones was associated with QRDR mutations. The MDR *S*. Typhi 4.3.1 subclade, which commonly also has reduced susceptibility to fluoroquinolones, is responsible for inter- and intra-continental spread, regional circulation, as well as local outbreaks in different parts of the world [18].

As the MDR S. Typhi population is increasing and spreading in different parts of the world, local/regional surveillance of MDR S. Typhi populations using both blood culture and WGS approach is important to report data on mechanisms responsible for resistance, for control and prevention of its spread, and to guide clinicians with available treatment options. However, in the African setting where typhoid fever is endemic, blood cultures for diagnostic confirmation and molecular characterization of *S*. Typhi are not performed routinely due to cost and infrastructure constraints [29]. Thus, there is paucity of data regarding the distribution of MDR S. Typhi genotypes and resistance mechanisms in Zanzibar, Tanzania, as in many other African countries. As part of a prospective study collecting blood-cultures from patients admitted with acute undifferentiated fever in Zanzibar, we here characterize S. Typhi phenotypically and genotypically to contribute to the knowledge base on typhoid fever in the region and to guide treatment and control.

Methods

Ethical statement

The research protocol was approved by the Zanzibar Medical Research and Ethical Committee (record no ZAMREC/0002/November/2014, renewal no ZAHREC/02/June/2019/41), and by the Regional Committee for Medical Research Ethics Health Region South East Norway (record no 2014/1940/REK South-East). Written informed consent was obtained from the patients or, in the case of children, from a parent or a responsible family member.

Study site

Mnazi Mmoja hospital (MMH), Zanzibar, Tanzania, is the main referral hospital of the Zanzibar Archipelago with a population estimated to about 1.4 million in 2015 [30]. The 544-bed hospital also offers primary and secondary health care for the residents of Zanzibar city with a population of about 600,000 and is a teaching hospital for the State University of Zanzibar.

Study population

Patients in the medical, pediatric and neonatal departments were enrolled in the study if they, either on admission or during their hospital stay, had fever (\geq 38.3°C in adults, \geq 38.5°C in children) or hypothermia (<36.0°C), or were otherwise suspected to have systemic bacterial infection as judged by the clinician. Demographic and clinical information was obtained. A total of 1037 of 1043 eligible patients with fever and/or suspicion of serious systemic infection admitted to Mnazi Mmoja Hospital, Zanzibar, were consecutively included from March 17, 2015, to October 4, 2016 (in one patient two blood cultures were taken. Six patients were excluded because the blood culture sample was lacking). In addition, the three accessible (of in total seven) *S*. Typhi strains isolated during the pilot study in the years 2012 to 2013 [31] were included in the analyses. The details of the methodology of the pilot study have been previously described [31]. Briefly, the pilot study was performed at the same departments of Mnazi Mmoja Hospital as the main study, whereas 469 participants (neonates, children and adults) presenting with signs of systemic infections were included. As for the present study, the pilot study included clinical data, and blood was collected for culture and susceptibility testing of isolated microbes [31].

Bacterial isolation and identification

Blood specimens were inoculated in BACTEC Myco/F lytic blood culturing vials (Becton Dickinson, Franklin Lakes, N.J.), one bottle per episode of febrile illness, incubated and analyzed as described earlier [31]. *Salmonella* Typhi isolates were identified by standard biochemical tests, the API 20E, VITEK 2 analysis using the identification cards for gram-negatives (both bioMérieux, Marcy-l'Etoile, France), and serogrouping by omnivalent A-67 and Vi-antigen (sifin diagnostics gmbh, Berlin, Germany). In the 61 isolates accessible for further analyses, the phenotypic identification was confirmed by whole genome sequencing.

Antimicrobial susceptibility testing

Antimicrobial susceptibility testing for ampicillin, cefotaxime, ceftazidime, ciprofloxacin, trimethoprim-sulfamethoxazole, and chloramphenicol was performed by disc diffusion technique (Oxoid, Basingstoke, United Kingdom), and, for azithromycin, by minimal inhibitory concentration (MIC) gradient test (Liofilchem, Roseto degli Abruzzi, Italy) as described in the EUCAST guidelines [32]. Fifty-nine of the 61 isolates (the remaining two isolates were not frozen) were sent to Bærum Hospital, Vestre Viken Hospital Trust, Norway, for quality control of identification and susceptibility testing as well as for further characterization. For susceptibility testing, the same antimicrobials and the same disc diffusion technique as in Zanzibar were used. In addition, ceftriaxone, ciprofloxacin and azithromycin were tested by the MIC gradient test (Liofilchem). The three isolates of the pilot study were analyzed at the Department of Clinical Science, University of Bergen, Norway, using the same techniques. The results of the susceptibility testing were interpreted by the S-I-R system according to the EUCAST guidelines v 12.0 [33]. For the testing for ciprofloxacin, the EUCAST criteria for *Salmonella* species have been applied, classifying the strains with MIC >0.06 mg/L as resistant.

Whole genome sequencing

Short read whole genome sequencing (WGS) was performed on 58 of the 61 *Salmonella* Typhi isolates (3 isolates not accessible) and on three *S*. Typhi strains of the pilot study [31]. For the 58 study strains, genomic DNA for the sequencing was extracted from single colonies using the Wizard genomic DNA Purification Kit (Promega, Madison, WI, USA) according to the

manufacturer's instructions. QIAxpert (QIAGEN, Valencia, CA, USA) was used for purity control. For ample library preparation TruSeq Nano reagents (Illumina, San Diego, CA, USA) were used. For the three strains of the pilot study, DNA isolation was performed using Invitrogen PureLink genomic DNA kit (Thermo Fisher Scientific, MA, USA) according to the manufacturer's recommendations. DNA library preparation was carried out using the standard protocol of the Nextera XT DNA Library Preparation Kit v. 3 (Illumina) up to the library amplification. Sequencing was performed on two runs of a MiSeq instrument (Illumina) with 300 bp paired end reads, according to the manufacturer's instructions. All 61 short-read sequences were *de novo* assembled using the Pathogenwatch online platform (https://pathogen.watch). Fifty-eight of these 61 short-read sequences, all from the main study, were, in addition, assembled against a reference genome using Bowtie v. 2.3.4.2 [34], Samtools v.1.7 [35] and BCFTools v. 1.9 [35,36], using the largest closed genome representative for one of the two lineages as reference.

Long read sequencing was performed on six selected *S*. Typhi isolates of the main study (representing isolates with different antimicrobial resistance patterns and from both years 2015 and 2016), as well as three *S*. Typhi strains from the pilot study (31). DNA isolation was performed using PureLink genomic DNA kit (Thermo Fisher Scientific) according to the manufacturer's recommendations. Nanopore sequencing was performed using the rapid barcoding kit (SQK-RBK004) and MinION sequencer (Oxford Nanopore Technologies Ltd., Oxford, United Kingdom), and Guppy v. 3.2.10 for basecalling [37]. Hybrid-assembly was carried out using Unicycler v.0.4.8 [38] running with Pilon v. 1.23 [39] and Racon v. 1.4.3 [40] for error correction and sequence polishing on 'normal' settings. Five of the six isolates were completely assembled resulting in one contig, while isolate 50-M123 was returned with 14 contigs. Medusa v. 1.6 [41] together with one closed assembly from the different lineages was used to sort the 14 contigs so that they matched the other *Salmonella* isolates as closely as possible. Manual editing of contigs were subsequently carried out on the 50-M123 isolate to align sequence start with the other isolates in the study.

Identification of resistance genes genotypes and sequence types

We used ResFinder (v4.1) with the default setting (90% threshold and 60% coverage) and MLST (v. 2.) with the default setting for the species, both from the Centre for Genomic Epidemiology CGE server (http://www.genomicepidemiology.org/), to identify acquired and chromosomal antimicrobial resistance determinants, and to assign sequence types, respectively. An online platform, Pathogenwatch (https://pathogen.watch) [42], was used to screen and assign *S*. Typhi genotypes. The Pathogenwatch tool uses GenoTyphi (code available at https://github. com/katholt/genotyphi) [42]. The GenoTyphi genotyping scheme divides the *Salmonella* Typhi population into four major lineages, and >75 different clades and subclades with the globally disseminated 4.3.1 (H58) subclade further delineated into lineages I and II (4.3.1.1 and 4.3.1.2) [15].

Plasmid analysis

The presence of plasmids was investigated using PlasmidFinder v. 2.1 with the default setting (Danish Technical University, Denmark: http://cge.cbs.dtu.dk/services/PlasmidFinder/).

Phylogenetic analysis

Conserved signature inserts phylogeny server (v. 1.4) (<u>https://cge.food.dtu.dk/services/</u> <u>CSIPhylogeny/</u>) was used to create a single nucleotide polymorphism (SNP)-based phylogenetic tree. All default values of the software were applied including minimum depth at SNP positions 10x, minimum relative depth at SNP positions 10%, minimum SNP quality 30. The first phylogenetic tree compared the relatedness of our *S*. Typhi isolates, in the tree we included all 61 *S*. Typhi from this study which underwent WGS. The *S*. Typhi strain with accession number ERL12960 of genotype 4.3.1.1, isolated in India in 2012 [43] was used as a reference genome for the phylogenetic tree. The second phylogenetic tree was created to compare the 61 *S*. Typhi isolates from this study which belong to 4.3.1 genotype and map their relationship with other strains to get more insights into the spread of the 4.3.1 genotype. We included 38 published WGS sequences of *S*. Typhi of the same genotype from different parts of the world [18,22,28,44], including 16 isolates from Tanzania, fourteen from Kenya, two from Zambia, five from India and one from Pakistan, from the years 2007 to 2017. Both phylogenetic trees were annotated using the Interactive Tree of Life (v. 5.6.3) [45].

Genetic environment for the MDR S. Typhi

The two representative *S*. Typhi from this study (ZNZ13L78 and ZNZ57M188) were annotated manually using a combination of RAST [46], Basic Local Alignment Search Tool (BLAST) (v. 2.11.0) [47], ResFinder (v. 4.1) [48] and MobileElementFinder (v. 1.0.3) from the CGE server and in SnapGene (v. 3.3.4) from GSL Biotech (available at <u>snapgene.com</u>). A comparative analysis using genoPlotr [49] was done for the chromosomal MDR gene segment of about 25kb of the two strains in this study and strain ERL12960.

Results

During the study period of about 19 months in 2015–2016, 1038 blood cultures of 1037 patients were obtained, and 161 (16%) of these had growth of pathogens. *S.* Typhi was the most common pathogen, found in 61 of the 161 (38%) positive blood cultures (including nine patients with two and two patients with three isolates), corresponding to 35% (61/174) of the pathogens recovered. Additionally, among seven *S.* Typhi detected during the eight months pilot study in 2012–2013 [31], three isolates were available for further analyses. During the main study, the average isolation rate was more than three isolates per month compared to one isolate per month during the pilot study.

Antimicrobial resistance pattern of S. Typhi

As shown in <u>Table 1</u>, MDR (defined as resistance towards ampicillin, trimethoprim-sulfamethoxazole and chloramphenicol) was observed in 98% of the isolates (63/64, including 60/61 from the main study and all three from the pilot study).

A total of 62 *S*. Typhi isolates (59 from the main study and three from the pilot study) were tested for ciprofloxacin resistance and 69% (43/62) showed low-level resistance with MIC values ranging from 0.125–0.25 mg/L. Of note, one of the *S*. Typhi isolates with low-level ciprofloxacin resistance was not MDR.

Results from disc diffusion showed all 64 S. Typhi isolates were susceptible to cefotaxime and ceftazidime. All tested isolates (fifty nine from the main study and three from the pilot study) were also susceptible to ceftriaxone by MIC gradient test with MIC values of 0.047–0.125 mg/L.

All 63 S. Typhi isolates were susceptible to azithromycin with MIC values 8-16 mg/L.

Whole genome sequencing results

Short read sequencing was performed on 61 S. Typhi isolates (fifty-eight from the main study and three from the pilot study). A total of 60 isolates (98.3%) which phenotypically were MDR

Antimicrobial agent	Number (%) of resistant isolates								
	Total	Genotype		Year					
		4.3.1.1	4.3.1.2	2012	2013	2015	2016		
Disc diffusion									
Ampicillin (n = 64)	63* (98%)	60 (100%)	0 (0%)	2 (100%)	1 (100%)	34 (100%)	24 (96%)		
Chloramphenicol (n = 64)	63* (98%)	60 (100%)	0 (0%)	2 (100%)	1 (100%)	34 (100%)	24 (96%)		
Trimethoprim-sulfamethoxazole ($n = 64$)	63* (98%)	60 (100%)	0 (0%)	2 (100%)	1 (100%)	34 (100%)	24 (96%)		
Cefotaxime (n = 64)	0* (0%)	0 (0%)	0 (0%)	0 (0%)	0 (0%)	0 (0%)	0 (0%)		
Ceftazidime (n = 64)	0* (0%)	0 (0%)	0 (0%)	0 (0%)	0 (0%)	0 (0%)	0 (0%)		
MIC gradient test									
Ceftriaxone (n = 62)	0* (0%)	0 (0%)	0 (0%)	0 (0%)	0 (0%)	0 (0%)	0 (0%)		
Ciprofloxacin (n = 62)	43* (69%)	41 (68%)	1 (100%)	0 (0%)	0 (0%)	22 (65%)	21 (84%)		
Azithromycin (n = 63)	0* (0%)	0 (0%)	0 (0%)	0 (0%)	0 (0%)	0 (0%)	0 (0%)		

Table 1. Antimicrobia	l resistance patterr	n for S. Typhi isolates.
-----------------------	----------------------	--------------------------

* including 3 isolates of the pilot study 2012–2013

https://doi.org/10.1371/journal.pntd.0012132.t001

carried antimicrobial resistance determinants associated with resistance to ampicillin (bla_{TEM-1}) , trimethoprim-sulfamethoxazole (sul1 and sul2 plus a dfrA7 gene) and chloramphenicol (catA1). In addition to MDR determinants, these isolates also carried AMR determinants conferring resistance towards aminoglycosides (aph(6)-Id and aph(3")-Ib). One S. Typhi which was not MDR phenotypically, did not carry any AMR determinants. Table 2 shows the distribution of AMR determinants including QRDR mutations of the three different AMR genotype profiles. Detailed information on the 61 genomes is listed in S1 Table.

Resistance toward fluoroquinolones was associated with a single QRDR mutation in 42/43 *S*. Typhi which showed phenotypic low-level resistance to ciprofloxacin (the remaining strain was not accessible to WGS). Forty-one strains had single *gyrA*-D87G QRDR mutation, while one strain which was not MDR had a different single *gyrA*-S83F mutation.

As shown in Table 2, all MDR S. Typhi isolates were assigned to the 4.3.1.1 genotype using the GenoTyphi scheme [15,42]. On the contrary, one S. Typhi isolate (ZNZ50M123) which was not MDR was assigned to the 4.3.1.2 genotype. Furthermore, all S. Typhi isolates belong to MLST sequence type ST1.

The genetic environment of resistance genes

Long-read sequencing was performed on six selected strains of the main study to investigate whether the AMR island is integrated in the chromosome or located on a plasmid. The gene contents of the five MDR *S*. Typhi isolates were highly similar, and the antimicrobial resistance determinants coding for MDR were located on a composite transposon integrated in the chromosome within a segment encompassed by insertion sequences and transposases. Fig 1 shows the structure of composite transposon of two *S*. Typhi isolates from this study ZNZ13L78 and ZNZ55M142 (with ENA accession numbers ERR11413875 and ERR11413524 respectively, for

Table 2. Antimicrobial resistance genes and QRDR mutations of the three AMR genotype profiles.

AMR genotype profile	Aminoglycoside	Beta lactam	Phenicol	Sulfamethoxazole, trimethoprim	QRDR mutations
MDR/cipR sub-lineage of 4.3.1.1	aph(6)-Id, aph(3")-Ib	bla _{TEM-1B}	catA1	sul1, sul2, dfrA7	gyrA: p.D87G
MDR/cipS sub-lineage of 4.3.1.1	aph(6)-Id, aph(3")-Ib	bla _{TEM-1B}	catA1	sul1, sul2, dfrA7	-
nonMDR/cipR 4.3.1.2 isolate	-	-	-	-	gyrA: p.S83F

https://doi.org/10.1371/journal.pntd.0012132.t002



Fig 1. The structure of MDR composite transposon. The red arrows show antimicrobial resistance genes, MDR genes are *sul1/sul2*, *dfrA7*, *bla*_{TEM-1} and *catA1*. Green arrows show insertion sequences (IS) or transposases and orange arrow show the replicons. BC04 is study isolate ZNZ13L78, BC06 is study isolate ZNZ55M142.

https://doi.org/10.1371/journal.pntd.0012132.g001

the short reads, and GenBank accession numbers SAMN35714885 and SAMN35714917 respectively, for the long reads) and ERL12960 as a comparison *S*. Typhi. All MDR *S*. Typhi isolates carried a composite transposon Tn6029 (encoding $bla_{\text{TEM-1B}}$, *sul2*, *aph*(6)-*Id* and *aph* (3")-*Ib*) which was inserted in Tn21 (carrying a class I integron encoding *dfrA* alleles in the gene cassette and *sul1*), which was in turn inserted within Tn9 (encoding *catA1*) with IS1 on both ends. No plasmid was found, but the MDR *S*. Typhi strains contained the IncQ1 plasmid replicon sequence (*repA* and *repC*) as shown in Fig 1. Furthermore, we detected chromosomal integration of the IS1 at the known site downstream of *cyaY* as previously described [22].

Phylogenetic analysis

Fig 2 is a whole genome SNP-based phylogenetic tree containing the 61 S. Typhi isolates from this study and one reference strain ERL12960. As shown in Fig 2, one isolate ZNZ50M123 (ENA accession numbers ERR11414360 and ERZ18316203, project *PRJEB59168*), highlighted in green belongs to genotype 4.3.1.2. This isolate has reduced susceptibility to ciprofloxacin (due to *gyrA*-S83F), and no acquired resistance genes. All other isolates were genotype 4.3.1.1, these were all MDR (due to *bla*_{TEM-1}, *catA1*, *sul1*, *sul2*, *dfrA7*). The majority of these belonged to a monophyletic clade (highlighted purple in tree) and had reduced susceptibility to ciprofloxacin (due to *gyrA*-D87G), herein referred as MDR/cipR. In contrast, the others (highlighted yellow in tree) including the older isolates from the pilot study, are fully sensitive to ciprofloxacin, and lack QRDR mutations referred as MDR/cipS. Within the study isolates, the SNP difference ranged between 0 and 41 (median 7.5), with the biggest SNP between the only isolate belonging to the 4.3.1.2 genotype (ZNZ50M123) and an isolate which is MDR/



Fig 2. Phylogenetic tree of S. Typhi isolates from this study using the reference genome ERL 12960 as midpoint root. The inner circle shows three clusters of our *S*. Typhi: purple highlighted is a monophyletic clade (MDR/cipR), these are MDR and have reduced susceptibility to ciprofloxacin due to *gyrA*-D87G; yellow highlighted are strains which are MDR but sensitive to ciprofloxacin (MDR/cipS); one strain in green highlight is not MDR and has low-level ciprofloxacin resistance due to single *gyrA*-S83F QRDR mutation. The second circle shows the year of isolation of *S*. Typhi; black represents *S*. Typhi isolated in 2016, grey in 2015, light grey and very light grey were isolated in 2013 and 2012 respectively. The next circle shows the season when *S*. Typhi were isolated, green represents isolates collected during the rainy season while orange highlighted *S*. Typhi were collected during the dry season. The outer circle shows the outcome of the patients; red represents *S*. Typhi from patients who died, blue represents *S*. Typhi from patients who survived while grey represents *S*. Typhi from patients with unknown outcome.

https://doi.org/10.1371/journal.pntd.0012132.g002

cipR. For the genomes belonging to the 4.3.1.1 genotype, the overall difference of all study isolates ranged between 0 and 27 SNPs (median 7), within MDR/cipR isolates between 0 and 16 SNPs (median 6), within MDR/cipS isolates between 3 and 17 SNPs (median 9.5). The MDR/ cipS isolates had a minimum difference of 11 SNPs to the closest neighbor of MDR/cipR isolates, with two of the pilot study isolates being closest related to MDR/cipR isolates. The only isolate of the genotype 4.3.1.2 (ZNZ50M123), had a difference between 25 and 41 SNPs (median 34) to the other study isolates (all belonging to 4.3.1.1 genotype).

In Fig 3, to provide context, the relationship of the 61 study strains to 38 strains from other publications including the reference strain ERL12960 is shown in a whole genome SNP-based phylogenetic tree. The minimum difference of the study isolates belonging to MDR/cipR group (genotype 4.3.1.1, MDR, and *gyrA*-D87G QRDR mutation) to isolates from other studies was 0 SNP in two isolates obtained in 2015 from travelers returning from Tanzania, both of which also were MDR and had identical *gyrA* point mutation, but no IncHI1 plasmid [28]. The 2nd most closely related isolate was also from Tanzania, and had 5 SNPs difference (year 2009, MDR, no IncHI1 plasmid, no FQ point mutation, ERR108659) [18]. The closest isolate





https://doi.org/10.1371/journal.pntd.0012132.g003

from Kenya had a difference of 17 SNPs. The isolates from Zambia had a difference of at least 15 SNPs, the minimum difference to isolates from Asia was 29 SNPs (India).

Isolates which are MDR/cipS had a minimum difference of 2 SNPs to isolates from East Africa (Tanzania, year 2009, with MDR, without the IncHI1 plasmid and without FQ point mutation) [18]. The closest isolate from Kenya had 11 SNPs difference. The difference to the Zambian isolates was at least 8 SNPs, and to the Asian isolates at least 22 SNPs (an isolate from India).

The non MDR genotype 4.3.1.2 isolate with the point mutation *gyrA*-S83F ZNZ50M123 had a minimum difference of 8 SNPs from another previously published Tanzanian isolate (year 2009, MDR, with IncHI1 plasmid and same *gyrA* point mutation, ERR108654) [18]. The second most closely related are four isolates from Tanzania with 10 SNPs distance (all from 2010, no MDR, no IncHI1 plasmid, all with the point mutation *gyrA*-S83F) [18]. The minimum distance to Kenyan isolates was 27 SNPs, to isolates from Zambia 29 SNPs, and to isolates from Asia (India) 21 SNPs.

All mentioned related external isolates belong to the ST1, as do the isolates from the current study.

Discussion

The study shows a high rate of multidrug resistance in *S*. Typhi from Zanzibar throughout the study period, concurring with findings from previous studies from Tanzania [25], neighboring Malawi, Kenya and Zambia and other African countries [19,22,44,50]. The prevalence of typhoid fever increased from 1% (7/469) of patients in the pilot study in 2012/2013 to 6% (61/1037) in the current study from 2015/2016. Correspondingly, *S*. Typhi accounted for 9% (7/79) and 35% (61/174) of pathogenic bacteria in blood cultures in the pilot and the current study, respectively, suggesting an outbreak in 2015/2016. Our findings support the report of an ongoing epidemic of MDR *S*. Typhi in Africa [18]. The high prevalence of MDR and reduced susceptibility to fluoroquinolones limit the treatment options of typhoid fever in Zanzibar considerably.

In this study all MDR *S*. Typhi belong to genotype 4.3.1.1. This genotype has been linked with inter- and intra-continental spread of MDR *S*. Typhi [18]. Therefore, it is likely that genotype 4.3.1 has been imported from South Asia in the last two decades to East African countries including Tanzania, Kenya and Uganda [44] resulting in further spread of the genotype and local outbreaks. This genotype is associated with resistance to all three former first line drugs; ampicillin, chloramphenicol and trimethoprim-sulfamethoxazole. The availability over the counter and widespread use of these antibiotics in the study setting may exert a selective pressure that contributes to maintaining the persistence of MDR *S*. Typhi in the region. The only *S*. Typhi isolate which is not MDR belongs to genotype 4.3.1.2. This genotype has been reported in Southeast Asian countries such as India, Nepal, and Bangladesh [18], neighboring countries Kenya and Uganda [16,44,51], as well as in Tanzania [44].

Ciprofloxacin and other fluoroquinolones are widely used for the treatment of typhoid fever as a consequence of the high prevalence of MDR [52]. This study found that a high proportion of *S*. Typhi genotype 4.3.1.1 had low-level ciprofloxacin resistance. This is alarming as treatment with fluoroquinolones may lead to treatment failure [53] and relapse. Increased doses and prolonged therapy may be effective but may fuel emergence of high-level resistance [8]. In a multi-country typhoid fever surveillance study in Africa, isolates from Tanzania showed a high percentage of MDR (89%, 8/9 isolates), but no fluoroquinolone resistance was found [25]. However, the fluoroquinolone resistance rates currently increase on the African continent [24], and overuse of fluoroquinolones is likely partly to blame [54]. Fortunately, high-level fluoroquinolone resistance was not detected in the study strains, but continued widespread use of fluoroquinolone may bring it on.

In the present study, fluoroquinolone resistance was conferred by chromosomal single point mutations leading to structural alterations in topoisomerases as DNA gyrase. Increasing numbers of point mutations are correlated with a cumulative increase in MIC values, with the simultaneous acquisition of at least three-point mutations resulting in high-level fluoroquino-lone resistance [24]. Changes in positions 83 and 87 of the *gyrA* gene are commonly reported point mutations [10,24]. All *S*. Typhi study isolates belonging to a monophyletic clade (MDR/ cipR) had a single *gyrA*-D87G QRDR mutation and were phenotypically expressing low-level fluoroquinolone resistance. To our knowledge, in East Africa the *gyrA*-D87G point mutation has only been described in two *S*. Typhi strains isolated from returning travelers from Tanzania [28].

In contrast, the only isolate without MDR (isolate ZNZ50M123, belonging to the 4.3.1.2 genotype) harbors a different single mutation (*gyrA*-S83F), another common mutation in fluoroquinolone resistant *S*. Typhi, including East African isolates [25]. Furthermore, all *S*. Typhi isolates in this study were sensitive to azithromycin which would remain the only effective oral treatment for typhoid fever. Globally, azithromycin resistant *S*. Typhi is rare, but increasingly

reported in South-East Asia, notably in Bangladesh [55]. Increased use of azithromycin will subsequently pose a risk of introducing and spreading azithromycin-resistant *S*. Typhi in the African region [55]. Third generation cephalosporins such as ceftriaxone are still effective in Zanzibar, unlike in Pakistan, that currently experiences a long-lasting outbreak with an extensive drug-resistant *S*. Typhi strain, resistant to ceftriaxone, ciprofloxacin, ampicillin, trimetho-prim-sulfamethoxazole, and chloramphenicol [52,55,56].

Earlier studies in Asia and some African countries have shown that MDR in S. Typhi is associated with the presence of an IncHI1 plasmid [18,44]. This means the main spread of antimicrobial resistance determinants has been horizontal transfer using a plasmid. Later studies have shown that MDR S. Typhi isolates from Zambia [22], Tanzania [44] and Asia [18,57] did not harbor plasmids associated with MDR determinants, suggesting that the genes conferring MDR have been incorporated in the chromosome of the bacteria. Concurring with previous findings [18,22,44], all MDR isolates in the present study carried a composite transposon integrated into the chromosome. Both the data from our study and from recent studies are compatible with spread of AMR through clonal expansion [18]. Previous studies suggest that the chromosomal location of the MDR determinants may confer a competitive advantage for the bacteria as it is less energy consuming compared to harboring a plasmid [28,44]. The integration of AMR determinants into the S. Typhi chromosome is worrying, as the chromosomal location reduces the likelihood of bacteria losing the antimicrobial resistance determinants [10]. The presumably low fitness cost associated with carriage of the MDR transposon [28] can provide a mechanism for sustained vertical transmission of MDR S. Typhi, even in the absence of selection pressure for the specific resistances [28].

In the phylogenetic comparison of the study isolates, monophyletic clade (MDR/cipR) isolates showed no SNP difference to the genotypes of *S*. Typhi isolated from two travelers returning to the United Kingdom from Tanzania during the same period [28]. They additionally share the same determinants coding for AMR including the identical *gyrA* point mutation, which, to our knowledge, has not been described in other *S*. Typhi isolates from East Africa. Isolates which are MDR/cipS, are closely related to other strains from Tanzania. Compared to MDR/cipR isolates, MDR/cipS has a smaller SNP difference to isolates from both Kenya and, especially, to those from Zambia. Assessing the results of the epidemiological and the antimicrobial resistance analyses, we speculate that the monophyletic clade (MDR/cipR) isolates may represent a new subtype and an outbreak strain, whereas MDR/cipS may be endemic.

The only non-MDR study strain is belonging to the genotype 4.3.1.2 and harboring a *gyrA*-S83F point mutation. It is closest related to a Tanzanian isolate from 2009 which had the same point mutation but differed by also harboring MDR and an IncHI1 plasmid. The second most closely related are four Tanzanian isolates from 2010 without MDR and without the IncHI1 plasmid, also with the same point mutation. In Asia, a decrease of MDR is associated with a corresponding decrease in carriage of IncHI1 plasmids [58], and we may speculate that the isolate from our study may also have lost the IncHI1 plasmid. This study isolate showed a smaller SNP difference to isolates from India than to isolates from neighboring Kenya or Zambia. This may support the earlier introduction and spread of a common ancestor from India to Tanzania.

In line with previous reports [42], the genetic findings in our study match the phenotypical results, emphasizing the potential utility of WGS for the prediction of AMR. The results are underlining that WGS is an important tool for surveillance of typhoid fever for uncovering outbreaks, and for understanding epidemiological relationships and the spread of antimicrobial resistance locally and globally. The high rate of MDR *S*. Typhi demonstrate the need of both antimicrobial stewardship for the treatment of suspected typhoid fever as well as surveillance. The study hospital is located on Unguja, the largest island of the Zanzibar Archipelago,

which has a multicultural population with historical links to mainland Tanzania, India and the Arabian Peninsula, in addition many international tourists visit the island. Extensive international travel may render the island vulnerable to the spread of resistant microbes [59], underlining the importance of continuous surveillance both locally and internationally.

Conclusions

We report high rate of MDR and low-level ciprofloxacin resistant *S*. Typhi genotype 4.3.1.1 circulating in Zanzibar. The findings support that this clade now prevails in East Africa [16,44,50], leaving few therapeutic options available for treatment of typhoid fever in the setting. Surveillance of the prevalence, spread and antimicrobial susceptibility of *S*. Typhi can guide treatment and control efforts.

Supporting information

S1 Table. Per-isolate information on AMR genotype profile, year, AMR determinants and accession numbers of genomes (submitted to ENA project PRJEB59168 and to GenBank BioProject PRJNA982791). (DOCX)

Acknowledgments

We thank all doctors, nurses, and all other staff at the Department of Internal Medicine and the Department of Pediatrics at Mnazi Mmoja Hospital, for their contribution to the study. We equally thank the technicians and all other staff of the Pathology Laboratory Department at Mnazi Mmoja Hospital who facilitated and contributed to the study. We are also grateful the support by Merriam Sundberg-Amargo and Harald Landa for their help in preparing the shipment of material from Norway to Zanzibar as well as by Inger Marie Brend Johansen, Mette Voldhaug and Tone Kofstad for the further analysis of the strains, all from the Department of Microbiology at Vestre Viken Hospital Trust, Norway.

The Illumina sequencing service of the study strains was provided by the Norwegian Sequencing Centre (www.sequencing.uio.no), a national technology platform hosted by Oslo University Hospital and the University of Oslo, supported by the "Research Council of Norway and the Southeastern Regional Health Authorities" and, for the three isolates of the pilot study, by Akershus University Hospital, Lørenskog, Norway.

The Illumina short read sequencing of three strains of the pilot study as well as the MinION sequencing (six strains of the main study and three strains of the pilot study) was provided by Akershus University Hospital, Lørenskog, Norway.

Jessin J. J. Peter, University of Tromsø, Norway, performed a first analysis of the WGS data. Thank you to R. S. Hendriksen, Technical University of Denmark, regarding the advice for the analysis by long read sequencing (MinION).

Author Contributions

Conceptualization: Annette Onken, Nina Langeland, Bjørn Blomberg.

Data curation: Annette Onken, Sabrina Moyo, Jon Bohlin, Bjørn Blomberg.

Formal analysis: Annette Onken, Sabrina Moyo, Jon Bohlin, Bjørn Blomberg.

Funding acquisition: Annette Onken, Nina Langeland, Kristine Mørch, Bjørn Blomberg.

- Investigation: Annette Onken, Sabrina Moyo, Mohammed Khamis Miraji, Jon Bohlin, Msafiri Marijani, Kibwana Omar Kibwana, Khamis Ali Abeid, Marianne Reimers, Bjørn Blomberg.
- **Methodology:** Annette Onken, Sabrina Moyo, Jon Bohlin, Fredrik Müller, Pål A. Jenum, Bjørn Blomberg.
- Project administration: Annette Onken, Msafiri Marijani, Bjørn Blomberg.
- **Resources:** Annette Onken, Sabrina Moyo, Mohammed Khamis Miraji, Jon Bohlin, Msafiri Marijani, Kibwana Omar Kibwana, Pål A. Jenum, Khamis Ali Abeid, Kristine Mørch, Bjørn Blomberg.
- Software: Annette Onken, Sabrina Moyo, Jon Bohlin, Bjørn Blomberg.
- Supervision: Annette Onken, Mohammed Khamis Miraji, Kibwana Omar Kibwana.
- Validation: Annette Onken, Sabrina Moyo, Jon Bohlin, Nina Langeland, Bjørn Blomberg.
- Visualization: Annette Onken, Sabrina Moyo, Bjørn Blomberg.
- Writing original draft: Annette Onken.
- Writing review & editing: Annette Onken, Sabrina Moyo, Mohammed Khamis Miraji, Jon Bohlin, Msafiri Marijani, Joel Manyahi, Kibwana Omar Kibwana, Fredrik Müller, Pål A. Jenum, Khamis Ali Abeid, Marianne Reimers, Nina Langeland, Kristine Mørch, Bjørn Blomberg.

References

- 1. Crump JA. Progress in Typhoid Fever Epidemiology. Clinical infectious diseases: an official publication of the Infectious Diseases Society of America. 2019; 68(Suppl 1):S4–s9.
- Global Burden of Disease Collaborative Network. GBD 2020 Cause and Risk Summaries: Typhoid fever—level 4 cause. Seattle, United States: Institute for Health Metrics and Evaluation (IHME). 2020 [cited 2022 07.02.2022]. Available from: https://www.healthdata.org/results/gbd_summaries/2019/ typhoid-fever-level-4-cause.
- Antillón M, Warren JL, Crawford FW, Weinberger DM, Kürüm E, Pak GD, et al. The burden of typhoid fever in low- and middle-income countries: A meta-regression approach. PLoS neglected tropical diseases. 2017; 11(2):e0005376. https://doi.org/10.1371/journal.pntd.0005376 PMID: 28241011
- The global burden of typhoid and paratyphoid fevers: a systematic analysis for the Global Burden of Disease Study 2017. Lancet Infect Dis. 2019; 19(4):369–81. <u>https://doi.org/10.1016/S1473-3099(18)</u> 30685-6 PMID: 30792131
- Parry CM, Hien TT, Dougan G, White NJ, Farrar JJ. Typhoid fever. The New England journal of medicine. 2002; 347(22):1770–82. https://doi.org/10.1056/NEJMra020201 PMID: 12456854
- Marks F, von Kalckreuth V, Aaby P, Adu-Sarkodie Y, El Tayeb MA, Ali M, et al. Incidence of invasive salmonella disease in sub-Saharan Africa: a multicentre population-based surveillance study. Lancet Glob Health. 2017; 5(3):e310–e23. https://doi.org/10.1016/S2214-109X(17)30022-0 PMID: 28193398
- Breiman RF, Cosmas L, Njuguna H, Audi A, Olack B, Ochieng JB, et al. Population-based incidence of typhoid fever in an urban informal settlement and a rural area in Kenya: implications for typhoid vaccine use in Africa. PloS one. 2012; 7(1):e29119. <u>https://doi.org/10.1371/journal.pone.0029119</u> PMID: 22276105
- Crump JA, Sjolund-Karlsson M, Gordon MA, Parry CM. Epidemiology, Clinical Presentation, Laboratory Diagnosis, Antimicrobial Resistance, and Antimicrobial Management of Invasive Salmonella Infections. Clinical microbiology reviews. 2015; 28(4):901–37. https://doi.org/10.1128/CMR.00002-15 PMID: 26180063
- 9. Bhutta ZA. Impact of age and drug resistance on mortality in typhoid fever. Arch Dis Child. 1996; 75 (3):214–7. https://doi.org/10.1136/adc.75.3.214 PMID: 8976660
- Dyson ZA, Klemm EJ, Palmer S, Dougan G. Antibiotic Resistance and Typhoid. Clinical infectious diseases: an official publication of the Infectious Diseases Society of America. 2019; 68(Supplement_2): S165–s70. https://doi.org/10.1093/cid/ciy1111 PMID: 30845331

- Le TA, Fabre L, Roumagnac P, Grimont PA, Scavizzi MR, Weill FX. Clonal expansion and microevolution of quinolone-resistant Salmonella enterica serotype typhi in Vietnam from 1996 to 2004. Journal of clinical microbiology. 2007; 45(11):3485–92. https://doi.org/10.1128/JCM.00948-07 PMID: 17728470
- Murdoch DA, Banatvaia N, Bone A, Shoismatulloev BI, Ward LR, Threlfall EJ. Epidemic ciprofloxacinresistant Salmonella typhi in Tajikistan. Lancet. 1998; 351(9099):339. <u>https://doi.org/10.1016/s0140-6736(05)78338-0 PMID: 9652619</u>
- Roumagnac P, Weill FX, Dolecek C, Baker S, Brisse S, Chinh NT, et al. Evolutionary history of Salmonella typhi. Science (New York, NY). 2006; 314(5803):1301–4. <u>https://doi.org/10.1126/science.</u> 1134933 PMID: 17124322
- Tacconelli E, Carrara E, Savoldi A, Harbarth S, Mendelson M, Monnet DL, et al. Discovery, research, and development of new antibiotics: the WHO priority list of antibiotic-resistant bacteria and tuberculosis. Lancet Infect Dis. 2018; 18(3):318–27. https://doi.org/10.1016/S1473-3099(17)30753-3 PMID: 29276051
- Wong VK, Baker S, Connor TR, Pickard D, Page AJ, Dave J, et al. An extended genotyping framework for Salmonella enterica serovar Typhi, the cause of human typhoid. Nature communications. 2016; 7:12827. https://doi.org/10.1038/ncomms12827 PMID: 27703135
- 16. Dyson ZA, Holt KE. Five years of GenoTyphi: updates to the global Salmonella Typhi genotyping framework. The Journal of infectious diseases. 2021.
- Carey ME, Dyson ZA, Ingle DJ, Amir A, Aworh MK, Chattaway MA, et al. Global diversity and antimicrobial resistance of typhoid fever pathogens: Insights from a meta-analysis of 13,000 Salmonella Typhi genomes. eLife. 2023; 12. https://doi.org/10.7554/eLife.85867 PMID: 37697804
- Wong VK, Baker S, Pickard DJ, Parkhill J, Page AJ, Feasey NA, et al. Phylogeographical analysis of the dominant multidrug-resistant H58 clade of Salmonella Typhi identifies inter- and intracontinental transmission events. Nature genetics. 2015; 47(6):632–9. https://doi.org/10.1038/ng.3281 PMID: 25961941
- Kariuki S, Revathi G, Kiiru J, Mengo DM, Mwituria J, Muyodi J, et al. Typhoid in Kenya is associated with a dominant multidrug-resistant Salmonella enterica serovar Typhi haplotype that is also widespread in Southeast Asia. Journal of clinical microbiology. 2010; 48(6):2171–6. <u>https://doi.org/10.1128/ JCM.01983-09 PMID: 20392916</u>
- Phan MD, Kidgell C, Nair S, Holt KE, Turner AK, Hinds J, et al. Variation in Salmonella enterica serovar typhi IncHI1 plasmids during the global spread of resistant typhoid fever. Antimicrob Agents Chemother. 2009; 53(2):716–27. https://doi.org/10.1128/AAC.00645-08 PMID: 19015365
- Holt KE, Phan MD, Baker S, Duy PT, Nga TV, Nair S, et al. Emergence of a globally dominant IncHI1 plasmid type associated with multiple drug resistant typhoid. PLoS neglected tropical diseases. 2011; 5 (7):e1245. https://doi.org/10.1371/journal.pntd.0001245 PMID: 21811646
- 22. Hendriksen RS, Leekitcharoenphon P, Lukjancenko O, Lukwesa-Musyani C, Tambatamba B, Mwaba J, et al. Genomic signature of multidrug-resistant Salmonella enterica serovar typhi isolates related to a massive outbreak in Zambia between 2010 and 2012. Journal of clinical microbiology. 2015; 53(1):262–72. https://doi.org/10.1128/JCM.02026-14 PMID: 25392358
- Ashton PM, Nair S, Dallman T, Rubino S, Rabsch W, Mwaigwisya S, et al. MinION nanopore sequencing identifies the position and structure of a bacterial antibiotic resistance island. Nature biotechnology. 2015; 33(3):296–300. https://doi.org/10.1038/nbt.3103 PMID: 25485618
- Britto CD, Wong VK, Dougan G, Pollard AJ. A systematic review of antimicrobial resistance in Salmonella enterica serovar Typhi, the etiological agent of typhoid. PLoS neglected tropical diseases. 2018; 12(10):e0006779. https://doi.org/10.1371/journal.pntd.0006779 PMID: 30307935
- 25. AI-Emran HM, Eibach D, Krumkamp R, Ali M, Baker S, Biggs HM, et al. A Multicountry Molecular Analysis of Salmonella enterica Serovar Typhi With Reduced Susceptibility to Ciprofloxacin in Sub-Saharan Africa. Clinical infectious diseases: an official publication of the Infectious Diseases Society of America. 2016; 62 Suppl 1:S42-6. https://doi.org/10.1093/cid/civ788 PMID: 26933020
- Hopkins KL, Davies RH, Threlfall EJ. Mechanisms of quinolone resistance in Escherichia coli and Salmonella: recent developments. International journal of antimicrobial agents. 2005; 25(5):358–73. https://doi.org/10.1016/j.ijantimicag.2005.02.006 PMID: 15848289
- Tadesse G, Tessema TS, Beyene G, Aseffa A. Molecular epidemiology of fluoroquinolone resistant Salmonella in Africa: A systematic review and meta-analysis. PloS one. 2018; 13(2):e0192575. <u>https://doi.org/10.1371/journal.pone.0192575 PMID: 29432492</u>
- Ingle DJ, Nair S, Hartman H, Ashton PM, Dyson ZA, Day M, et al. Informal genomic surveillance of regional distribution of Salmonella Typhi genotypes and antimicrobial resistance via returning travellers. PLoS neglected tropical diseases. 2019; 13(9):e0007620. <u>https://doi.org/10.1371/journal.pntd.0007620</u> PMID: 31513580

- Ombelet S, Ronat JB, Walsh T, Yansouni CP, Cox J, Vlieghe E, et al. Clinical bacteriology in lowresource settings: today's solutions. Lancet Infect Dis. 2018; 18(8):e248–e58. https://doi.org/10.1016/ S1473-3099(18)30093-8 PMID: 29519767
- Tanzania NBoS. Tanzania in figures 2015 2016 [29.08.2022]. Available from: https://www.nbs.go.tz/ index.php/en/tanzania-in-figures.
- Onken A, Said AK, Jorstad M, Jenum PA, Blomberg B. Prevalence and Antimicrobial Resistance of Microbes Causing Bloodstream Infections in Unguja, Zanzibar. PloS one. 2015; 10(12):e0145632. https://doi.org/10.1371/journal.pone.0145632 PMID: 26700032
- EUCAST. [29.08.2022]. Available from: https://www.eucast.org/ast_of_bacteria/previous_versions_of_ documents/.
- EUCAST clinical breakpoints (bacteria) v 12.0: EUCAST; 2022 [updated 01.01.2022. Available from: https://www.eucast.org/clinical_breakpoints.
- Langmead B, Salzberg SL. Fast gapped-read alignment with Bowtie 2. Nat Methods. 2012; 9(4):357–9. https://doi.org/10.1038/nmeth.1923 PMID: 22388286
- Li H, Handsaker B, Wysoker A, Fennell T, Ruan J, Homer N, et al. The Sequence Alignment/Map format and SAMtools. Bioinformatics (Oxford, England). 2009; 25(16):2078–9. <u>https://doi.org/10.1093/</u> bioinformatics/btp352 PMID: 19505943
- Li H. A statistical framework for SNP calling, mutation discovery, association mapping and population genetical parameter estimation from sequencing data. Bioinformatics (Oxford, England). 2011; 27 (21):2987–93. https://doi.org/10.1093/bioinformatics/btr509 PMID: 21903627
- Wick RR, Judd LM, Holt KE. Performance of neural network basecalling tools for Oxford Nanopore sequencing. Genome Biol. 2019; 20(1):129. https://doi.org/10.1186/s13059-019-1727-y PMID: 31234903
- Wick RR, Judd LM, Gorrie CL, Holt KE. Unicycler: Resolving bacterial genome assemblies from short and long sequencing reads. PLoS Comput Biol. 2017; 13(6):e1005595. https://doi.org/10.1371/journal. pcbi.1005595 PMID: 28594827
- Walker BJ, Abeel T, Shea T, Priest M, Abouelliel A, Sakthikumar S, et al. Pilon: an integrated tool for comprehensive microbial variant detection and genome assembly improvement. PloS one. 2014; 9(11): e112963. https://doi.org/10.1371/journal.pone.0112963 PMID: 25409509
- Vaser R, Sović I, Nagarajan N, Šikić M. Fast and accurate de novo genome assembly from long uncorrected reads. Genome Res. 2017; 27(5):737–46. <u>https://doi.org/10.1101/gr.214270.116</u> PMID: 28100585
- Bosi E, Donati B, Galardini M, Brunetti S, Sagot MF, Lió P, et al. MeDuSa: a multi-draft based scaffolder. Bioinformatics (Oxford, England). 2015; 31(15):2443–51. https://doi.org/10.1093/bioinformatics/ btv171 PMID: 25810435
- 42. Argimón S, Yeats CA, Goater RJ, Abudahab K, Taylor B, Underwood A, et al. A global resource for genomic predictions of antimicrobial resistance and surveillance of Salmonella Typhi at pathogenwatch. Nature communications. 2021; 12(1):2879. https://doi.org/10.1038/s41467-021-23091-2 PMID: 34001879
- 43. Sikorski MJ, Hazen TH, Desai SN, Nimarota-Brown S, Tupua S, Sialeipata M, et al. Persistence of Rare Salmonella Typhi Genotypes Susceptible to First-Line Antibiotics in the Remote Islands of Samoa. mBio. 2022; 13(5):e0192022. https://doi.org/10.1128/mbio.01920-22 PMID: 36094088
- Park SE, Pham DT, Boinett C, Wong VK, Pak GD, Panzner U, et al. The phylogeography and incidence of multi-drug resistant typhoid fever in sub-Saharan Africa. Nature communications. 2018; 9(1):5094. https://doi.org/10.1038/s41467-018-07370-z PMID: 30504848
- Letunic I, Bork P. Interactive Tree Of Life (iTOL) v5: an online tool for phylogenetic tree display and annotation. Nucleic Acids Res. 2021; 49(W1):W293–w6. https://doi.org/10.1093/nar/gkab301 PMID: 33885785
- Aziz RK, Bartels D, Best AA, DeJongh M, Disz T, Edwards RA, et al. The RAST Server: rapid annotations using subsystems technology. BMC genomics. 2008; 9:75. https://doi.org/10.1186/1471-2164-9-75 PMID: 18261238
- Altschul SF, Gish W, Miller W, Myers EW, Lipman DJ. Basic local alignment search tool. J Mol Biol. 1990; 215(3):403–10. https://doi.org/10.1016/S0022-2836(05)80360-2 PMID: 2231712
- **48.** ResFinder, Danish Technical University, Denmark [20.10.2022]. Available from: https://cge.food.dtu. dk/services/ResFinder/.
- Guy L, Kultima JR, Andersson SG. genoPlotR: comparative gene and genome visualization in R. Bioinformatics (Oxford, England). 2010; 26(18):2334–5. https://doi.org/10.1093/bioinformatics/btq413 PMID: 20624783

- Feasey NA, Gaskell K, Wong V, Msefula C, Selemani G, Kumwenda S, et al. Rapid emergence of multidrug resistant, H58-lineage Salmonella typhi in Blantyre, Malawi. PLoS neglected tropical diseases. 2015; 9(4):e0003748. https://doi.org/10.1371/journal.pntd.0003748 PMID: 25909750
- Kariuki S, Dyson ZA, Mbae C, Ngetich R, Kavai SM, Wairimu C, et al. Multiple introductions of multidrug-resistant typhoid associated with acute infection and asymptomatic carriage, Kenya. eLife. 2021; 10. https://doi.org/10.7554/eLife.67852 PMID: 34515028
- Marchello CS, Carr SD, Crump JA. A Systematic Review on Antimicrobial Resistance among Salmonella Typhi Worldwide. Am J Trop Med Hyg. 2020; 103(6):2518–27. <u>https://doi.org/10.4269/ajtmh.20-0258</u> PMID: 32996447
- Threlfall EJ, Ward LR, Skinner JA, Smith HR, Lacey S. Ciprofloxacin-resistant Salmonella typhi and treatment failure. Lancet. 1999; 353(9164):1590–1. https://doi.org/10.1016/s0140-6736(99)01001-6 PMID: 10334265
- 54. Mashe T, Leekitcharoenphon P, Mtapuri-Zinyowera S, Kingsley RA, Robertson V, Tarupiwa A, et al. Salmonella enterica serovar Typhi H58 clone has been endemic in Zimbabwe from 2012 to 2019. The Journal of antimicrobial chemotherapy. 2021; 76(5):1160–7. https://doi.org/10.1093/jac/dkaa519 PMID: 33347558
- 55. Saha S, Sajib MSI, Garrett D, Qamar FN. Antimicrobial Resistance in Typhoidal Salmonella: Around the World in 3 Days. Clinical infectious diseases: an official publication of the Infectious Diseases Society of America. 2020; 71(Suppl 2):S91–s5.
- 56. Klemm EJ, Shakoor S, Page AJ, Qamar FN, Judge K, Saeed DK, et al. Emergence of an Extensively Drug-Resistant Salmonella enterica Serovar Typhi Clone Harboring a Promiscuous Plasmid Encoding Resistance to Fluoroquinolones and Third-Generation Cephalosporins. mBio. 2018; 9(1). https://doi. org/10.1128/mBio.00105-18 PMID: 29463654
- Chiou CS, Alam M, Kuo JC, Liu YY, Wang PJ. Chromosome-mediated multidrug resistance in Salmonella enterica serovar Typhi. Antimicrob Agents Chemother. 2015; 59(1):721–3. <u>https://doi.org/10.1128/AAC.04081-14</u> PMID: 25367917
- da Silva KE, Tanmoy AM, Pragasam AK, Iqbal J, Sajib MSI, Mutreja A, et al. The international and intercontinental spread and expansion of antimicrobial-resistant Salmonella Typhi: a genomic epidemiology study. Lancet Microbe. 2022; 3(8):e567–e77. https://doi.org/10.1016/S2666-5247(22)00093-3 PMID: 35750070
- Woerther PL, Andremont A, Kantele A. Travel-acquired ESBL-producing Enterobacteriaceae: impact of colonization at individual and community level. J Travel Med. 2017; 24(suppl_1):S29–s34. <u>https://doi.org/10.1093/jtm/taw101</u> PMID: 28520999